Author's response to reviews

Title: Managing hospital doctors and their practice: what can we learn about human resource and performance management from non-healthcare organisations?

Authors:

- Timothy M Trebble Dr (tim.trebble@porthosp.nhs.uk)
- Nicola Heyworth Mrs (nickyheyworth@gmail.com)
- Nicholas Clarke Professor (n.r.clarke@soton.ac.uk)
- Timothy Powell Mr (tim.powell@porthosp.nhs.uk)
- Peter M Hockey Dr (peter.hockey@wessexdeanery.nhs.uk)

Version: 2
Date: 20 December 2013

Author's response to reviews:

Sir

that you for your request for revisions relating to the formatting of the manuscript that I have hopefully addressed

Tim Trebble